The Planning Act 2008

The Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 Regulation 5(2)(d)

**Thurrock Flexible Generation Plant** 

November May 2020

**Book of Reference (Parts 1-5)** 

**Document Reference: 4.3** 

**Author: Ardent Management Limited** 

Version 32





### **Thurrock Flexible Generation Plant**

# **BOOK OF REFERENCE**

## Contents

Land within the administrative boundaries of Thurrock Borough Council

Part 1	1
Part 2a	
Part 2b	
Part 3	
Part 4	
Part 5	143 <del>139143</del>

#### Introduction

- 1.1 This Book of Reference ("BoR") accompanies the proposed Development Consent Order ("the Order") for the Thurrock Flexible Generation Plant ("the authorised project"), as required by Regulation 5(2)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009 ("the Regulations"). It describes all the land, and identifies all the interests, affected by the Order.
- 1.2 The land described in this BoR, which is to be subject to powers of compulsory acquisition and/or temporary possession, is required to deliver the works comprising the authorised project which are described in detail in Part 1 of Schedule 1 of the Order and shown on the works plans.
- 1.3 Every parcel of land that is affected is identified on a plot by plot basis and a unique number has been ascribed to each plot.

  The plots are shown on the land plans which accompany the Order.
- 1.4 This BoR is comprised of five Parts, in accordance with Regulation 7(1) of the Regulations, as follows:
  - (a) Part 1 contains the names and addresses of those who own, lease, occupy or have another interest in the land that will be affected by the authorised project and the rights contained in the Order;
  - (b) Part 2 has been divided into two parts (Parts 2a and Part 2b) to aid clarity of presentation of those who may be entitled to make a relevant claim. Part 2a contains the names and addresses of those who have an interest in the Order land and may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised project. They are listed in this part in their capacity as category 3 persons per section 57 of the 2008 Act. Those persons also appear in part 1 of this Book of Reference as category 2 interests. Part 2b contains the names and addresses of those without an interest in the Order land, who may be entitled to claim compensation for loss resulting from the implementation of the Order and use of the authorised project;

- (c) Part 3 contains the names and addresses of those entitled to enjoy easements or other private rights which it is proposed may be extinguished, suspended or interfered with in connection with the authorised project, pursuant, Order;
- (d) Part 4 identifies plots in which there is a Crown interest that will be affected by the authorised project and the rights contained in the Order (there is no Crown Land in this case and this section is deliberately blank); and
- (e) Part 5 identifies plots which constitute "special category land" for the purposes of section 132 of the Planning Act 2008 that will be affected by the authorised project and the rights contained in the Order. (In this case common land, which is identified as such on the land plans).
- 1.5 The Order seeks powers to compulsorily acquire land and new rights (both temporary and permanent) for the purposes of the construction and operation of the authorised project.
- 1.6 Some of the plots identified in Part 1 of the BoR will be subject, acquisition of permanent rights (including restrictive covenants) pursuant to Article 22 (Compulsory acquisition of rights) and Schedule 5 (Land in which only new rights etc. may be required) of the Order. These plots are shown coloured blue on the land plans.

A number of the plots identified in Part 1 of the BoR will be subject to freehold acquisition pursuant to Article 18 (Compulsory acquisition of land) of the Order. These plots are shown coloured pink on the land plans.

A number of the plots identified in Part 1 of the BoR will be subject to the acquisition of permanent rights and imposition of restrictive covenants pursuant to Article 22 and Schedule 5, the rights which may be acquired in each plot are set out in Schedule 5 of the Order. These plots are shown coloured blue on the land plans.

Plots that are subject to powers of temporary possession only, such as for the purpose of access and diversion of a public rights of way during construction are listed in Schedule 6 of the Order and shown coloured green on the land plans.

The land shown coloured pink and blue on the land plans will also be subject to powers of temporary possession for the purpose of carrying out the authorised project (by virtue of Article 28 (Temporary use of land for carrying out the authorised project)).

By virtue of Article 29 (Temporary use of land for maintaining authorised project) of the Order, any land within the Order limits which is reasonably required for the purpose of maintaining the authorised project may be entered and/or temporarily possessed.

### Part 1

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
01/01	New rights over 387.05 square metres of land being arable field, west of Fort Road and south of railway, Tilbury.  Freehold title  EX987997 EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH Diana Mary Cole Chorry Orchard Farm Conways Road Orsett Grays	None	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)  Bloor Homes Limited Ashby Road
		RM16-3EL  James Andrew Cole  Mill House  Muckingford Road  West Tilbury  Tilbury  RM18-8TP		RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)  C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of unilateral notice)None

1

4.3: Book of Reference Parts 1-5

Commented [AP1]: Registered title has been updated since submission.

A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
 A person is within Category 2 if the applicant, after making diligent inquiry knows that the person (a) is interested in the land
 (b) has power (i) to sell or convey the land
 (ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
01/02	New rights over 106.19 square metres of land being arable field, west of Fort Road and south of railway, Tilbury.  Unregistered	Unknown  Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (inrespect of subsoil) Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)	None	Unknown  Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (inrespect of subsoil) Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)	None
01/03	New rights over 74.98 square metres of land being track, west of Fort Road and south of railway, Tilbury.	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett	None	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	Unregistered	Grays RM16-3EL (in respect of subsoil)		Grays RM16 3EL (in respect of subsoil)	
		James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18-8TP (in respect of subsoil)		James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)	
		Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of subsoil)		Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of subsoil)	
01/04	New rights over 8735.56 square metres of land being public road (Fort Road), private road (Power Station Approach Road, Fort Road), north of Tilbury Power Station, Tilbury.	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH	None	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH	AWG Land Holdings Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon Cambridgeshire
	Freehold title EX932756	Thurrock Borough Council Civic Centre New Road Grays		Thurrock Borough Council Civic Centre New Road Grays	PE29 6XÜ (in respect of rights) Ingrebourne Valley Limited

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		Essex RM17 6SL (in respect of adopted highway)		Essex RM17 6SL (in respect of adopted highway)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)  Telewest Communications (South East) Limited 500 Brook Drive Reading RG2 6UU (in respect of apparatus)  Vodafone Limited Vodafone House The Connection Newbury Berkshire	Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)  Eastern Power Networks plc Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of deed of gran

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				RG14 2FN (in respect of apparatus) SSE Enterprise Telecoms Limited One Forbury Place 43 Forbury Road Reading RG1 3JH (in respect of apparatus)	
01/05	Permanent acquisition of 58.58 square metres of land being arable field and drain, east of Fort Road, Tilbury.  Freehold title EX966447	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)
01/06	Permanent acquisition of 432.31 square metres of land being arable field, east of Fort Road, Tilbury.	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury	Thurrock Power Limited 1st Floor 145 Kensington Church Street

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	Freehold title EX76273 Leasehold title EX102229	Grays RM16 3EL	Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	London W8 7LP (as beneficiary)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)
01/07	Permanent acquisition of 115779.30 square metres of land being arable field,	Diana Mary Cole Cherry Orchard Farm Conways Road	Jeremy Godsmark Finnis Mill House Muckingford Road	Jeremy Godsmark Finnis Mill House Muckingford Road	Thurrock Power Limited 1st Floor 145 Kensington Church

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power
(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	east of Fort Road, Tilbury.  Freehold title EX76273  Leasehold title EX102229	Orsett Grays RM16 3EL	West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of rights for maintanence)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)
01/08	Permanent acquisition of 3253.04 square metres of	Diana Mary Cole Cherry Orchard Farm	Jeremy Godsmark Finnis Mill House	Jeremy Godsmark Finnis Mill House	Thurrock Power Limited 1st Floor

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power
(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	land being arable field and drain, east of Fort Road, Tilbury.  Freehold Title EX76273  Leasehold Title EX102229	Conways Road Orsett Grays RM16 3EL	Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Tilbury Essex RM18 8TP	Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	145 Kensington Church Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of rights for maintanence)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)
01/09	New rights over 3054.05	RWE Generation (UK) plc	National Grid Electricity	National Grid Electricity	Ingrebourne Valley Limited

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	square metres of land being grassland, overhead transmission lines and pylons, north of Tilbury Power Station, Tilbury. Freehold title EX639032 Leasehold title EX487728	Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	Transmission plc 1-3 Strand London WC2N 5EH	Transmission plc 1-3 Strand London WC2N 5EH  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
01/10	New rights over 3185.91 square metres of land being hardstanding at Tilbury Power Substation, Tilbury.  Freehold title EX639032 Leasehold title EX487728	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH	National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH  Vodafone Limited Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)  BT Openreach Limited 81 Newgate Street London	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				EC1A 7AJ (in respect of apparatus)  Virgin Media Limited 500 Brook Drive Reading RG2 6UU (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
01/11	New rights over 4698.52 square metres of land being private road (unnamed), north of Tilbury Power Station, Tilbury.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB  Vodafone Limited Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
01/12	Permanent acquisition of 6622.41 square metres of land being grassland south of Parsonage Common, Tilbury.  Unregistered	Unknown	None	Unknown  Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	None
01/13	Permanent acquisition of 4400.60 square metres of land being grassland south of Parsonage Common, Tilbury.  Freehold title EX966447	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary) Bloor Homes Limited Ashby Road

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		West Tilbury Tilbury RM18 8TP		West Tilbury Tilbury RM18 8TP	Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)  C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of unilateral notice and rights of common)  Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  Ann Louise Cole 44 St Johns Road Writtle Chelmsford

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					Essex CM1 3EB (in respect of rights of common)  Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					(in respect of rights of common)
01/14	New rights over 855.70 square metres of land being arable field, overhead transmission lines and drain, Parsonage Common, Tilbury.  Freehold title EX966447	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				(in respect of apparatus)	
01/15	New rights over 968.66 square metres of land being hardstanding at Tilbury Power Substation, Tilbury.  Freehold title EX639032 Leasehold title EX487728	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH	National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH  Vodafone Limited Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)  BT Openreach Limited 81 Newgate Street London EC1A 7AJ (in respect of apparatus)  Virgin Media Limited 500 Brook Drive Reading RG2 6UU (in respect of apparatus)  UK Power Networks Limited Newington House	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power
(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
01/16	Permanent acquisition of 100695.37 square metres of land being grassland, drains, pylons and overhead transmission lines, south of Parsonage Common, Tilbury.  Freehold title EX966447	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP  Vodafone Limited Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)  UK Power Networks Limited Newington House	Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)  Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  Ann Louise Cole 44 St Johns Road Writtle Chelmsford Essex CM1 3EB (in respect of rights of

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)  Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)	common)  Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					Port of Tilbury London Limited Leslie Ford House Tilbury RM18 7EH (as beneficiary)  C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
01/17	Permanent acquisition of 131997.60 square metres of land being arable field, drains, pylons and overhead transmission lines, south of Parsonage Common, Tilbury.  Freehold title EX76275, EX95363 Leasehold title EX102231	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
			Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Vodafone Limited Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH	

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Occupiers or Reputed Occupiers  (in respect of apparatus)  Anglian Water Services	Category 2 <sup>2</sup> owners
Anglian Water Services	
Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)	
Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)  Sheila Elizabeth Hodson	None
	Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)  Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power
(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		Conways Road Orsett Grays RM16 3EL (in respect of subsoil)		Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
01/19	Permanent acquisition of 11445.30 square metres of land being grassland, hardstanding, private road (unnamed), trees, shrubbery and overhead transmission lines, south east of Tilbury Power Substation, Tilbury.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)  Vodafone Limited Vodafone House The Connection	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)  Hanson Quarry Products Europe Limited Hanson House 14 Castle Hill Maidenhead SL6 4JJ (in respect of rights)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners	Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners	
				Newbury Berkshire RG14 2FN (in respect of apparatus) BT Openreach Limited 81 Newgate Street London EC1A 7AJ (in respect of apparatus)  Virgin Media Limited 500 Brook Drive Reading RG2 6UU (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  Telewest Communications (South East) Limited 500 Brook Drive Reading RG2 6UU		

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				(in respect of apparatus)	
01/20	Permanent acquisition of 27967.48 49805.08  square metres of land being grassland, shrubbery, pylons and overhead transmission lines, east of Walton Common, Tilbury.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  Vodafone Limited Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

4.3: Book of Reference Parts 1-5

Commented [AP2]: Amendment to boundary has reduced the

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
01/21	New rights over 98.22 square metres of land being hardstanding, north east of Tilbury Power Station, Tilbury.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
01/22	Permanent acquisition of 1060.13 square metres of land being hardstanding at Tilbury Power Station, Tilbury.  Freehold title EX639032 Leasehold title EX826359	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH	National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
01/23	Permanent acquisition of 337.44 square metres of land being grassland, drain, trees and overheard transmission lines, east of Walton Common, Tilbury.	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett	None	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	Unregistered	Grays RM16 3EL (in respect of subsoil)		Grays RM16 3EL (in respect of subsoil)	
		Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil) RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (in respect of subsoil)		Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (in respect of subsoil)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  National Grid Electricity	
				SE1 6NP (in respect of apparatus)	

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				London WC2N 5EH (in respect of apparatus)	
01/24	Permanent acquisition of 2986.39 square metres of land being grassland, shrubbery, overhead transmission lines and drains on the south side of the railway, south of Parsonage Shaw, Tilbury.  *Unregistered**  **Transpiration of Parsonage Shaw of P	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)	None	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury Tilbury Tilbury RM18 8TP (in respect of subsoil)  Vodafone Limited	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)	
01/25	New rights over 297.70 square metres of land being arable field, drains, south east of Parsonage Common, Tilbury.  Freehold title EX246891	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL James Andrew Cole	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL James Andrew Cole	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power
(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		Mill House Muckingford Road West Tilbury Tilbury RM18 8TP		Mill House Muckingford Road West Tilbury Tilbury RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
01/26	Permanent acquisition of 24.70 square metres of land being grassland and drain, east of Walton Common, Tilbury.  Unregistered	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		(in respect of subsoil)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (in respect of subsoil)		(in respect of subsoil)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (in respect of subsoil)	
01/26a	Permanent acquisition of 216.48 square metres of land being grassland and drain, east of Walton Common, Tilbury.  Unregistered	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon	None	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		SN5 6PB (in respect of subsoil)		SN5 6PB (in respect of subsoil)	
01/27	Permanent acquisition of 10460.75 square metres of land being arable field, south east of Walton Common, Tilbury.  Freehold title EX246891	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
01/28	Permanent acquisition of 1992.46 square metres of land being grassland east of Tilbury Power Station. Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)
01/29	Permanent acquisition of 50.19 square metres of land being track, north east of Tilbury Power Station,	Unknown  Diana Mary Cole Cherry Orchard Farm	None	Unknown  Diana Mary Cole Cherry Orchard Farm	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

(a) is interested in the land
(b) has power
(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	Tilbury.  Unregistered	Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (in respect of subsoil)		Conways Road Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)  RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB (in respect of subsoil)	
01/30	New rights over 7490.22 square metres of land being grassland, hardstanding, private road (unnamed), trees, shrubbery and overhead transmission lines, south east of Tilbury Power Substation, Tilbury.	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB  National Grid Electricity Transmission plc 1-3 Strand London	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary) Hanson Quarry Products

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	Freehold title EX639032			WC2N 5EH (in respect of apparatus)  Vodafone Limited Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)  BT Openreach Limited 81 Newgate Street London EC1A 7AJ (in respect of apparatus)  Virgin Media Limited 500 Brook Drive Reading RG2 6UU (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	Europe Limited Hanson House 14 Castle Hill Maidenhead SL6 4JJ (in respect of rights)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				Telewest Communications (South East) Limited 500 Brook Drive Reading RG2 6UU (in respect of apparatus)	
01/31	New rights over 95.22 square metres of land being hardstanding and private road, east of Walton Common, Tilbury.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB  Vodafone Limited Vodafone House The Connection Newbury Berkshire RG14 2FN (in respect of apparatus)	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
01/32	Permanent acquisition of 52.12 square metres of land being hardstanding, east of Walton Common, Tilbury.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehall Way Swindon SN5 6PB Vodafone Limited Vodafone House The Connection	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				Newbury Berkshire RG14 2FN (in respect of apparatus)	
02/01	New rights over 143.18 square metres of land being arable field and drain, north west of Parsonage Common, Tilbury.  Freehold title EX966447	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)  C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  Ann Louise Cole 44 St Johns Road Writtle Chelmsford Essex CM1 3EB (in respect of rights of common)  Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					Conways Road Orsett Grays RM16 3EL (in respect of rights of common) Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
02/02	New rights over 6964.99 square metres of land being arable field and drain, west of Parsonage Common, Tilbury.  Freehold title EX76273 Leasehold title EX102229	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power
(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
			Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
02/03	Permanent acquisition of 10596.91 square metres of land being arable field and drain, east of Fort Road, Tilbury.  Freehold title EX76273  Leasehold title EX102229	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
			RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	RM17 6SL (in respect of rights for maintanence)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)
02/04	New rights over 1539.37 square metres of land being Parsonage Common, south of Cooper Shaw Road, Tilbury  Freehold title EX966447	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		RM18 8TP		RM18 8TP	Derbyshire DE12 7JP (as beneficiary)  C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  Ann Louise Cole 44 St Johns Road Writtle Chelmsford Essex CM1 3EB (in respect of rights of

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					common)  Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
02/05	Permanent acquisition of 78.25 square metres of land being trees at Parsonage Common, West Tilbury.  Freehold title EX95363	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)
02/06	Permanent acquisition of 45291.69 square metres of land being arable field, track, drains, pylons and overhead transmission lines, south of Parsonage Common, West Tilbury.	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL Sheila Elizabeth Hodson	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	Freehold title EX76275, EX95363 Leasehold title EX102231	Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)	
02/07	Permanent acquisition of 52640.80 square metres of land being arable field, shrubbery, drains, pylons, overhead transmission lines at Parsonage Shaw, West Tilbury.  Freehold title EX95363, EX418108 Leasehold title EX102231	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Tilbury Essex RM18 8TP	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)  Anglian Water Services Limited Lancaster House Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)	
02/08	New rights over 4.48 square metres of land being grassland and drain at Parsonage Common, West Tilbury.	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road	None	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	Unregistered	Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)		Orsett Grays RM16 3EL (in respect of subsoil)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)	
02/09	New rights over 25.27 square metres of land being grassland and drain at Parsonage Common, West Tilbury.  Unregistered Leasehold title EX102231	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP Diana Mary Cole Cherry Orchard Farm	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP Diana Mary Cole Cherry Orchard Farm	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of subsoil) Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)	Conways Road Orsett Grays RM16 3EL Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Conways Road Orsett Grays RM16 3EL Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
02/10	New rights over 233.72 square metres of land being highway verge and hardstanding (Cooper Shaw Road), west of Parsonage Common, West Tilbury  **Unregistered**	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)	None	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)		Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of subsoil)	
02/11	New rights over 253.28 square metres of land being shrubbery at Parsonage Shaw, West Tilbury.  Freehold title EX95363, EX418108 Leasehold title EX102231	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

	Description of Land	Category 1 <sup>1</sup> owners			
Number on Plan		Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
02/12	Permanent acquisition of 2376.65 square metres of land being grassland, track and drain, south of Parsonage Shaw, West Tilbury.  Unregistered	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	None	Unknown  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of way)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of rights of way)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				(in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
02/13	New rights over 60035.74 square metres of land being arable field, track, pylons and overhead transmission lines, west of Parsonage Shaw, West Tilbury.  Freehold title EX246891	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				SE1 6NP (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)  Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)	
02/14	Permanent acquisition of 38353.62 square metres of land being arable field, track and drains, west of Parsonage Shaw, West Tilbury.  Freehold title EX246891	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		RM18 8TP		RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London WC2N 5EH (in respect of apparatus)  Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)	London W8 7LP (as beneficiary)
03/01a	New rights over 7893.09 square metres of land being arable land, pylons and overhead transmission	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

or Reputed Category 2 <sup>2</sup> owners
Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  House vark Bridge  of apparatus) id Electricity on plc  of apparatus) ter Services
rror N

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				Huntingdon PE29 6XU (in respect of apparatus)	
03/01	New rights over 34276.76 square metres of land being arable land, pylons and overhead transmission lines and drain, south of Station Road, Tilbury.  Freehold title EX246891	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  National Grid Electricity Transmission plc 1-3 Strand London	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				WC2N 5EH (in respect of apparatus)  Anglian Water Services Limited Lancaster House Lancaster Way Ermine Business Park Huntingdon PE29 6XU (in respect of apparatus)	
03/02	New rights over 5690.37 square metres of land being arable land and drain, south of Station Road, Tilbury.  Freehold title EX246891	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	None	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
03/03	New rights over 1168.17 square metres of land being public footpath	Diana Mary Cole Cherry Orchard Farm Conways Road	None	Diana Mary Cole Cherry Orchard Farm Conways Road	Thurrock Power Limited 1st Floor 145 Kensington Church

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	(Footpath 146), trees and drain, south west of Station Road, East Tilbury.  Freehold title EX972747	Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP		Orsett Grays RM16 3EL  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of public rights of way)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	Street London W8 7LP (as beneficiary)  Cogent Land LLP 33 Margaret Street London W1G 0JD (in respect of an option)  International Power Limited Level 20 25 Canada Square London E14 5LQ (in respect of deed)  Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  Ann Louise Cole 44 St Johns Road Writtle

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					Chelmsford Essex CM1 3EB (in respect of rights of common)  Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					RM18 8TP (in respect of rights of common)  C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
03/04	New rights over 14664.93 square metres of land being agricultural field, trees and shrubbery, west of Gravelpit Farm, Tilbury.  Freehold title EX246889	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR  Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury Tilbury RM18 8QR	None	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR  UK Power Networks	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)  International Power Limited Level 20 25 Canada Square London E14 5LQ (in respect of deed)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
03/05	New rights over 1319.91 square metres of land being public highway (Station Road), West Tilbury.  Unregistered	Unknown  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of adopted highway)  Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR (in respect of subsoil)  Rita Maureen Mott Goshem's Farm	None	Unknown  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of adopted highway)  Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR (in respect of subsoil)  Rita Maureen Mott Goshem's Farm	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		Station Road East Tilbury Tilbury RM18 8QR (in respect of subsoil)		Station Road East Tilbury Tilbury RM18 8QR (in respect of subsoil)	
03/06	New rights over 15027.72 square metres of land being arable field, grassland, trees and shrubbery at Gravelpit Farm, Tilbury.  Freehold title EX246889 and EX763684	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR  Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR	None	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR  Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

	Description of Land	Category 1 <sup>1</sup> owners			
Number on Plan		Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
03/07	Temporary rights over 1167.65 square metres of land being private road (unnamed), west of Goshem's Farm, East Tilbury.  Unregistered	Ghislane Rossi Buckland Station Road East Tilbury Tilbury RM18 8QR	None	Ghislane Rossi Buckland Station Road East Tilbury Tilbury RM18 8QR	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR (in respect of rights of access)  Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR (in respect of rights of access)  David Glenn Mott Norrsken Station Road East Tilbury Tilbury RM18 8QR (in respect of rights of access)  David Glenn Mott Norrsken Station Road East Tilbury Tilbury RM18 8QR (in respect of rights of access)  Karen Misra Norrsken

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					Station Road East Tilbury Tilbury RM18 8QR (in respect of rights of access)  Exel Logistics Property Limited Ocean House The Ring Bracknell Berkshire RG12 1AN (in respect of rights of access)
03/08	New rights over 1656.20 square metres of land being public highway (Station Road), West Tilbury.  Unregistered	Unknown Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of adopted highway) Diana Mary Cole Cherry Orchard Farm Conways Road	None	Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of adopted highway)  Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		Orsett		RM16 3EL	
		Grays		(in respect of subsoil)	
		RM16 3EL			
		(in respect of subsoil)		Sue Cole	
				Mill House	
		Sue Cole		Muckingford Road	
		Mill House		West Tilbury	
		Muckingford Road		Tilbury	
		West Tilbury		Essex	
		Tilbury		RM18 8TP	
		Essex		(in respect of subsoil)	
		RM18 8TP			
		(in respect of subsoil)		James Andrew Cole	
		, ,		Mill House	
		James Andrew Cole		Muckingford Road	
		Mill House		West Tilbury	
		Muckingford Road		Tilbury	
		West Tilbury		RM18 8TP	
		Tilbury		(in respect of subsoil)	
		RM18 8TP			
		(in respect of subsoil)		BT Openreach Limited	
				81 Newgate Street	
		Melville Hamilton Lowe		London	
		Mott		EC1A 7AJ	
		Goshem's Farm		(in respect of apparatus)	
		Station Road			
		East Tilbury		Essex and Suffolk Water	
		Tilbury		Limited	
		Essex		Northumbria House	
		RM18 8QR		Abbey Road	
		(in respect of subsoil)		Pity Me	

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR (in respect of subsoil)		Durham DH1 5FJ (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
03/09	New rights over 21164.86 square metres of land being grassland and arable field at Goshem's Farm, East Tilbury.  Freehold title EX246889	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR	None	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)
03/10	Temporary rights over 832.72 square metres of	Unknown	None	Unknown	None

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	land being public highway (Station Road), West Tilbury. <i>Unregistered</i>	Thurrock Borough Council Civic Centre New Road Grays Essex		Thurrock Borough Council Civic Centre New Road Grays Essex	
		RM17 6SL (in respect of adopted highway)		RM17 6SL (in respect of adopted highway)	
		Thomas Crane Pleasant View Love Lane East Tilbury Village Essex RM18 8QJ (in respect of subsoil)		Thomas Crane Pleasant View Love Lane East Tilbury Village Essex RM18 8QJ (in respect of subsoil)	
		Eunice Crane Pleasant View Love Lane East Tilbury Village Essex RM18 8QJ (in respect of subsoil)		Eunice Crane Pleasant View Love Lane East Tilbury Village Essex RM18 8QJ (in respect of subsoil)	
		Paul Anthony Lennon Willows Station Road East Tilbury Essex RM18 8QR		Paul Anthony Lennon Willows Station Road East Tilbury Essex RM18 8QR	

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		(in respect of subsoil)  Anita Elizabeth Lennon Willows Station Road East Tilbury Tilbury RM18 8QR (in respect of subsoil)  Raymond Louis Osborne Addison Princess Margaret Road East Tilbury Essex RM18 8PA (in respect of subsoil)  Karl Alexander Osborne Buxton Princess Margaret Road East Tilbury Essex RM18 8PA (in respect of subsoil)  Mark Raymond Osborne Rosenholme Princess Margaret Road East Tilbury Essex RM18 8PA (in respect of subsoil)		(in respect of subsoil)  Anita Elizabeth Lennon Willows Station Road East Tilbury Tilbury RM18 8QR (in respect of subsoil)  Raymond Louis Osborne Addison Princess Margaret Road East Tilbury Essex RM18 8PA (in respect of subsoil)  Karl Alexander Osborne Buxton Princess Margaret Road East Tilbury Essex RM18 8PA (in respect of subsoil)  Mark Raymond Osborne Rosenholme Princess Margaret Road East Tilbury Essex RM18 8PA (in respect of subsoil)	

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
		RM18 8PA (in respect of subsoil)		RM18 8PA (in respect of subsoil)	
		Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR (in respect of subsoil)		Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR (in respect of subsoil)	
		Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR (in respect of subsoil)		Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR (in respect of subsoil)	
				BT Openreach Limited 81 Newgate Street London EC1A 7AJ (in respect of apparatus)	
				Cadent Gas Limited Ashbrook Court Prologis Park Central Boulevard	

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				Coventry CV7 8PE (in respect of apparatus)  Essex and Suffolk Water Limited Northumbria House Abbey Road Pity Me Durham DH1 5FJ (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)	
03/11	Permanent acquisition of 22827.83 square metres of land being agricultural and arable fields, trees and shrubbery at Goshem's Farm, East Tilbury.  Freehold title EX246889	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR	None	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary) International Power Limited Level 20 25 Canada Square

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners			
	Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
			Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR		Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR  National Grid Gas plc 1-3 Strand London WC2N 5EH (in respect of apparatus)	London E14 5LQ (in respect of deed)
					Cadent Gas Limited Ashbroek Court Prologis Park Central Boulevard Coventry CV7 8PE (in respect of apparatus)  UK Power Networks	
-	04/01	New rights over 7181.29	RWE Generation (UK) plc	None	Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  RWE Generation (UK) plc	Port of Tilbury London

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power
(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

4.3: Book of Reference Parts 1-5

**Commented [AP3]:** Confirmation email from National Grid Gas that they have apparatus in this parcel of land, not Cadent Gas.

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
	square metres of land being private access road (unnamed), hardstanding, grassland and shrubbery, east of Tilbury Power Station, Tilbury.  Freehold title EX639032	Windmill Hill Business Park Whitehill Way Swindon SN5 6PB		Windmill Hill Business Park Whitehill Way Swindon SN5 6PB  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of public rights of way)  Essex County Council County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public rights of way)  National Grid Electricity Transmission Plc 1 - 3 Strand London WC2N 5EH (in respect of apparatus)  UK Power Networks	Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus) Telewest Communications (South East) Limited 500 Brook Drive Reading RG2 6UU (in respect of apparatus)	
04/02	Permanent acquisition of 15213.50 square metres of land being public footpath (Footpath 146), grassland, trees, shrubbery and hardstanding, east of Tilbury Power Station, Tilbury.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of public rights of way)  Essex County Council	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)  Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
				County Hall Market Road Chelmsford Essex CM1 1QH (in respect of public rights of way)  National Grid Electricity Transmission Plc 1 - 3 Strand London WC2N 5EH (in respect of apparatus)  UK Power Networks Limited Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of apparatus)  Telewest Communications (South East) Limited 500 Brook Drive Reading RG2 6UU (in respect of apparatus)	notice and beneficiary)

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

			Category 1 <sup>1</sup> owners		
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
04/03	New rights over 11346.49 square metres of land being private acces road (unnamed), quarryland and grassland, east of Tilbury Power Station.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)  Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
04/04	Number Not Used	-	-	-	-
04/05	Permanent acquisition of 7692.92 square metres of land being grassland, east of Tilbury Power Station, Tilbury.  Freehold title EX639032	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB	None	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary) Ingrebourne Valley Limited Cecil House Foster Street

<sup>1.</sup> A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.

2. A person is within Category 2 if the applicant, after making diligent inquiry knows that the person(a) is interested in the land
(b) has power(i) to sell or convey the land
(ii) to release the land; see section 57 (2) of the Planning Act 2008

		Category 1 <sup>1</sup> owners			
Number on Plan	Description of Land	Freehold or Reputed Freehold Owners	Lessees or Tenants or Reputed Lessees or Tenants	Occupiers or Reputed Occupiers	Category 2 <sup>2</sup> owners
					Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
04/06	Permanent acquisition of 145.19 square metres of land being private access road (unnamed), quarryland and grassland, east of Tilbury Power Station.  Unregistered	Unknown  RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (in respect of subsoil)	None	Unknown  RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (in respect of subsoil)	None

A person is within Category 1 if the applicant, after making diligent inquiry knows that the person is an owner, lessee, tenant (whatever the tenancy period) or occupier of the land; see section 57 (1) of the Planning Act 2008.
 A person is within Category 2 if the applicant, after making diligent inquiry knows that the person (a) is interested in the land
 (b) has power (i) to sell or convey the land
 (ii) to release the land; see section 57 (2) of the Planning Act 2008

## Part 2a

	Category 3 <sup>3</sup>		
Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973	
New rights over 387.05	Port of Tilbury London Limited	None	
square metres of land	Leslie Ford House		
being arable field, west of	<del>Tilbury</del>		
Fort Road and south of	Essex		
railway, Tilbury.	RM18 7EH		
	(in respect of unilateral notice and beneficiary)		
Freehold title FX966447			
, recircia a are 2,1000 r.m	Bloor Homes Limited		
	Ashby Road		
	<del>Measham</del>		
	Swadlincote		
	<del>Derbyshire</del>		
	<del>DE12 7JP</del>		
	<del>(as beneficiary)</del>		
	C H Cole and Sons		
	Mill House		
	Muckingford Road		
	West Tilbury		
	Tilbury		
	Essex		
	RM18-8TP		
	(in respect of unilateral notice)		
New rights over 8735 56	AWG Land Holdings Limited	None	
Road), private road (Power			
	being arable field, west of Fort Road and south of railway, Tilbury.	New rights over 387.05 square metres of land being arable field, west of Fort Read and south of railway, Tilbury.  Freehold title EX966117  Bloor Homes Limited Ashby Read Measham Swadlincete Derbyshire DE12 7JP (as beneficiary)  CH Cole and Sons Mill House Muckingford Read West Tilbury Tilbury Essex RM18 8TP (in respect of unilateral notice)  New rights over 8735.56 square metres of land being public road (Fort Road), private road (Power	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

4.3: Book of Reference Parts 1-5

**Commented [AP4]:** Title register has beenupdated since submsiion, no longer any part 2a interests.

Number		С	ategory 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	Fort Road), north of Tilbury Power Station, Tilbury.	Cambridgeshire PE29 6XU (in respect of rights)	
	Freehold title EX932756	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)  Eastern Power Networks plc Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of deed of grant)	
01/05	Permanent acquisition of 58.58 square metres of land being arable field and drain, east of Fort Road, Tilbury.  Freehold title EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

	C	Category 3 <sup>3</sup>
Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	(as beneficiary)	
Permanent acquisition of 432.31 square metres of land being arable field, east of Fort Road, Tilbury.  Freehold title EX76273 Leasehold title EX102229	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)	None
Permanent acquisition of 115779.30 square metres of land being arable field, east of Fort Road, Tilbury.  Freehold title EX76273 Leasehold title EX102229	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of rights for maintanence)	None
	432.31 square metres of land being arable field, east of Fort Road, Tilbury.  Freehold title EX76273 Leasehold title EX102229  Permanent acquisition of 115779.30 square metres of land being arable field, east of Fort Road, Tilbury.  Freehold title EX76273	Claimant under section 10 of the Compulsory Purchase Act 1965  (as beneficiary)  Permanent acquisition of 432.31 square metres of land being arable field, east of Fort Road, Tilbury.  Freehold title EX76273 Leasehold title EX102229  Permanent acquisition of 115779.30 square metres of land being arable field, east of Fort Road, Tilbury.  Freehold title EX76273 Leasehold title EX7628  Reasehold title EX76273 Leasehold title EX76273 Leasehold title EX7628  Reasehold title EX7628  Reasehold title EX76273 Leasehold title EX7628  Reasehold title EX7628  Reasehold title EX7628  Reasehold title EX76273 Leasehold title EX7628  Reasehold title EX7628  Reas

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		С	Category 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)	
01/08	Permanent acquisition of 3253.04 square metres of land being arable field and drain, east of Fort Road, Tilbury.  Freehold Title EX76273 Leasehold Title EX102229	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of rights for maintanence)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)	None
01/09	New rights over 3054.05 square metres of land being grassland, overhead transmission lines and	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		C	Category 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	pylons, north of Tilbury Power Station, Tilbury. Freehold title EX639032 Leasehold title EX487728	Essex CM17 9HY (in respect of unilateral notice and beneficiary)	
01/10	New rights over 3185.91 square metres of land being hardstanding at Tilbury Power Substation, Tilbury.  Freehold title EX639032 Leasehold title EX487728	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	None
01/11	New rights over 4698.52 square metres of land being private road (unnamed), north of Tilbury Power Station, Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	None
01/13	Permanent acquisition of 4400.60 square metres of land being grassland south of Parsonage Common, Tilbury.  Freehold title EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary) Bloor Homes Limited Ashby Road	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Ca	tegory 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Measham	1919
		Swadlincote	
		Derbyshire	
		DE12 7JP	
		(as beneficiary)	
		C H Cole and Sons	
		Mill House	
		Muckingford Road	
		West Tilbury	
		Tilbury	
		Essex	
		RM18 8TP	
		(in respect of unilateral notice and rights of common)	
		Jeremy Godsmark Finnis	
		Mill House	
		Muckingford Road	
		West Tilbury	
		Tilbury	
		Essex	
		RM18 8TP	
		(in respect of rights of common)	
		Ann Louise Cole	
		44 St Johns Road	
		Writtle	
		Chelmsford	
		Essex	
		CM1 3EB	
		(in respect of rights of common)	
		Kathryn Ksenia Finnis	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number	Description of Land	Ca	ategory 3 <sup>3</sup>
on Plan		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)	
01/14	New rights over 855.70 square metres of land being arable field, overhead transmission lines and drain, Parsonage Common, Tilbury	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	Freehold title EX966447	Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)	
01/15	New rights over 968.66 square metres of land being hardstanding at Tilbury Power Substation, Tilbury.  Freehold title EX639032 Leasehold title EX487728	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	None
01/16	Permanent acquisition of 100695.37 square metres of land being grassland, drains, pylons and overhead transmission lines, south of Parsonage Common, Tilbury.  Freehold title EX966447	Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)  Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Ca	ategory 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		(in respect of rights of common)	
		Ann Louise Cole	
		44 St Johns Road	
		Writtle	
		Chelmsford	
		Essex	
		CM1 3EB	
		(in respect of rights of common)	
		Kathryn Ksenia Finnis	
		Wyfields Farm	
		Blackbush Lane	
		Horndon on the Hill	
		Stanford-Le-Hope Essex	
		SS17 8PT	
		(in respect of rights of common)	
		Sheila Elizabeth Hodson	
		Cherry Orchard Farm	
		Conways Road Orsett	
		Grays	
		RM16 3EL	
		(in respect of rights of common)	
		Sue Cole	
		Mill House	
		Muckingford Road	
		West Tilbury	
		Tilbury	
		Essex	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number	Description of Land	Category 3 <sup>3</sup>	
on Plan		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		RM18 8TP (in respect of rights of common)  Port of Tilbury London Limited Leslie Ford House Tilbury RM18 7EH (as beneficiary)  C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)	
01/17	Permanent acquisition of 131997.60 square metres of land being arable field, drains, pylons and overhead transmission lines, south of Parsonage Common, Tilbury.  Freehold title EX76275, EX95363 Leasehold title EX102231	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
01/19	Permanent acquisition of 11445.30 square metres of land being grassland, hardstanding, private road (unnamed), trees, shrubbery and overhead transmission lines, south east of Tilbury Power Substation, Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary) Hanson Quarry Products Europe Limited Hanson House 14 Castle Hill Maidenhead SL6 4JJ (in respect of rights)	None
01/20	Permanent acquisition of 27967.4849805.08 square metres of land being grassland, pylons and overhead transmission lines, east of Walton Common, Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	None
01/21	New rights over 98.22 square metres of land being hardstanding, north east of Tilbury Power Station, Tilbury.	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number	Description of Land	Category 3 <sup>3</sup>	
on Plan		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	Freehold title EX639032	(in respect of unilateral notice and beneficiary)	
01/22	Permanent acquisition of 1060.13 square metres of land being hardstanding at Tilbury Power Station, Tilbury.  Freehold title EX639032 Leasehold title EX826359	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	None
01/25	New rights over 297.70 square metres of land being arable field, drains, south east of Parsonage Common, Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary) Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)	None
01/27	Permanent acquisition of 10460.75 square metres of land being arable field, south east of Walton Common, Tilbury.	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	Freehold title EX246891	Thurrock Power Limited 1st Floor	
		145 Kensington Church Street London	
		W8 7LP (as beneficiary)	
01/28	Permanent acquisition of 1992.46 square metres of land being grassland east of Tilbury Power Station.	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH	None
	Freehold title EX639032	(in respect of unilateral notice and beneficiary)	
01/30	New rights over 7490.22 square metres of land being grassland, hardstanding, private road (unnamed), trees, shrubbery and overhead transmission lines, south east of Tilbury Power Substation, Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary) Hanson Quarry Products Europe Limited Hanson House 14 Castle Hill	
01/31	New rights over 95.22	Maidenhead SL6 4JJ (in respect of rights)  Ingrebourne Valley Limited	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	square metres of land being hardstanding and private road, east of Walton Common, Tilbury. Freehold title EX639032	Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	
01/32	Permanent acquisition of 52.12 square metres of land being hardstanding, east of Walton Common, Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	
02/01	New rights over 143.18 square metres of land being arable field and drain, north west of Parsonage Common, Tilbury.  Freehold title EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)  C H Cole and Sons Mill House	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  Ann Louise Cole 44 St Johns Road Writtle Chelmsford Essex CM1 3EB (in respect of rights of common)  Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Ca	ategory 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common) Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)	
02/02	New rights over 6964.99 square metres of land being arable field and drain, west of Parsonage Common, Tilbury.  Freehold title EX76273 Leasehold title EX102229	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)	None
02/03	Permanent acquisition of	Thurrock Power Limited	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	10596.91 square metres of land being arable field and drain, east of Fort Road, Tilbury.  Freehold title EX76273 Leasehold title EX102229	1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of rights for maintanence)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)	
02/04	New rights over 1539.37 square metres of land being Parsonage Common, south of Cooper Shaw Road, Tilbury  Freehold title EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Ca	ategory 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Derbyshire DE12 7JP (as beneficiary)	
		C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)	
		Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)	
		Ann Louise Cole 44 St Johns Road Writtle Chelmsford Essex CM1 3EB (in respect of rights of common)	
		Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		c	ategory 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)	
02/05	Permanent acquisition of 78.25 square metres of land being trees at Parsonage Common, West Tilbury.  Freehold title EX95363	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)	None
02/06	Permanent acquisition of	Thurrock Power Limited	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

<sup>4.3:</sup> Book of Reference Parts 1-5

Number	Description of Land	Category 3 <sup>3</sup>	
on Plan		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	45291.69 square metres of land being arable field, track, drains, pylons and overhead transmission lines, south of Parsonage Common, West Tilbury.  Freehold title EX76275, EX95363 Leasehold title EX102231	1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)	
02/07	Permanent acquisition of 52640.80 square metres of land being arable field, shrubbery, drains, pylons, overhead transmission lines at Parsonage Shaw, West Tilbury.  Freehold title EX95363, EX418108 Leasehold title EX102231	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)	None
02/09	New rights over 25.27 square metres of land being grassland and drain at Parsonage Common, West Tilbury.	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number on Plan	Description of Land	Category 3 <sup>3</sup>	
		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	Unregistered Leasehold title EX102231	(as beneficiary)	
02/11	New rights over 253.28 square metres of land being shrubbery at Parsonage Shaw, West Tilbury.  Freehold title EX95363, EX418108 Leasehold title EX102231	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)	None
02/12	Permanent acquisition of 2376.65 square metres of land being grassland, track and drain, south of Parsonage Shaw, West Tilbury.  Unregistered	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of way)  James Andrew Cole Mill House Muckingford Road West Tilbury RM18 8TP	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		(in respect of rights of way)	
02/13	New rights over 60035.74 square metres of land being arable field, track, drains, pylons and overhead transmission lines, west of Parsonage Shaw, West Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)	None
02/14	Permanent acquisition of 38353.62 square metres of land being arable field, track and drains, west of Parsonage Shaw, West Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
03/01a	New rights over 7893.09 square metres of land being arable land, pylons and overhead transmission lines and drain, south of Station Road, Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary) Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)	None
03/01	New rights over 34276.76 square metres of land being arable land, pylons and overhead transmission lines and drain, south of Station Road, Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)	None
03/02	New rights over 5690.37 square metres of land	RWE Generation (UK) plc Windmill Hill Business Park	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	being arable land and drain, south of Station Road, Tilbury.	Whitehill Way Swindon Wiltshire SN5 6PB	
	Freehold title EX246891	(as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)	
03/03	New rights over 1168.17 square metres of land being public footpath (Footpath 146), trees and drain, south west of Station Road, East Tilbury.  Freehold title EX972747	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Cogent Land LLP 33 Margaret Street London W1G 0JD (in respect of an option)	None
		International Power Limited Level 20 25 Canada Square London E14 5LQ (in respect of deed)	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

<sup>4.3:</sup> Book of Reference Parts 1-5

Number		Ca	tegory 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury	
		Essex RM18 8TP (in respect of rights of common)	
		Ann Louise Cole 44 St Johns Road Writtle Chelmsford Essex CM1 3EB (in respect of rights of common)	
		Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)	
	ithin Catagon 2 if the Applicant thinks that if the	Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number	Description of Land	Category 3 <sup>3</sup>	
on Plan		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)  C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)	
03/04	New rights over 14664.93 square metres of land being agricultural field, trees and shrubbery, west of Gravelpit Farm, Tilbury.  Freehold title EX246889	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)  International Power Limited Level 20 25 Canada Square London E14 5LQ	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number	Description of Land	Category 3 <sup>3</sup>	
on Plan		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		(in respect of deed)	
03/06	New rights over 15027.72 square metres of land being arable field, grassland, trees and shrubbery at Gravelpit Farm, Tilbury.  Freehold title EX246889 and EX763684	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)	None
03/07	Temporary rights over 1167.65 square metres of land being private road (unnamed), west of Goshem's Farm, East Tilbury.  Unregistered	Melville Hamilton Lowe Mott Goshem's Farm Station Road East Tilbury Tilbury Essex RM18 8QR (in respect of rights of access)  Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury Tilbury RM18 8QR (in respect of rights of access)  David Glenn Mott Norrsken Station Road	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number	Description of Land	C	ategory 3 <sup>3</sup>
on Plan		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		East Tilbury Tilbury RM18 8QR (in respect of rights of access)  Karen Misra Norrsken Station Road East Tilbury Tilbury RM18 8QR (in respect of rights of access)  Exel Logistics Property Limited Ocean House The Ring Bracknell Berkshire RG12 1AN (in respect of rights of access)	
03/09	New rights over 21164.86 square metres of land being grassland and arable field at Goshem's Farm, East Tilbury.  Freehold title EX246889	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)	None
03/11	Permanent acquisition of 22827.83 square metres of land being agricultural and	Cogent Land LLP 33 Margaret Street London	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		Category 3 <sup>3</sup>	
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	arable fields, trees and shrubbery at Goshem's Farm, East Tilbury.	W1G 0JD (as beneficiary) International Power Limited	
	Freehold title EX246889	Level 20 25 Canada Square London E14 5LQ (in respect of deed)	
04/01	New rights over 7181.29 square metres of land being private access road (unnamed), hardstanding, grassland and shrubbery, east of Tilbury Power Station, Tilbury.  Freehold title EX639032	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	None
04/02	Permanent acquisition of 15213.50 square metres of land being public footpath (Footpath 146), grassland, trees, shrubbery and hardstanding, east of Tilbury Power Station,	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number		С	Category 3 <sup>3</sup>
on Plan	Description of Land	Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
	Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	
04/03	New rights over 11346.49 square metres of land being private acces road (unnamed), quarryland and grassland, east of Tilbury Power Station.  Freehold title EX639032	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary) Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	None
04/05	Permanent acquisition of 7692.92 square metres of land being grassland, east of Tilbury Power Station, Tilbury.  Freehold title EX639032	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary) Ingrebourne Valley Limited Cecil House	None

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Number on Plan	Description of Land	Category 3 <sup>3</sup>	
		Claimant under section 10 of the Compulsory Purchase Act 1965	Claimant under Part 1 of the Land Compensation Act 1973
		Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Part 2b

Category 31 Potential Claimant under Part 1 of the Land Compensation Act 1973		
Description of Land	Contact Details	
1 West Cottages, Low Street Lane, East Tilbury, RM18 8RA	Grace Elizabeth Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	
2 West Cottages, Low Street Lane, East Tilbury, RM18 8RA	Grace Elizabeth Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	
3 West Cottages, Low Street Lane, East Tilbury, RM18 8RA	Grace Elizabeth Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	
Polwicks, Church Road, West Tilbury, RM18 8QU	James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Category 31 Potential Claimant under Part 1 of the Land Compensation Act 1973		
Description of Land	Contact Details	
Polwicks Farm House, Church Road, West Tilbury, RM18 8QU	John Royston Lawrence Polwicks Farm House Church Road West Tilbury Essex RM18 8QU  Patricia Ann Lawrence Polwicks Farm House Church Road West Tilbury Essex RM18 8QU	
Highhouse, Highhouse Lane, West Tilbury, RM18 8TL	James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Category 31 Potential Claimant under Part 1 of the Land Compensation Act 1973		
Description of Land	Contact Details	
1 Condovers Cottages Church Road, East Tilbury, RM18 8QX	Diana Mary Finnis Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Jeremy Paul Godsmark Finnis Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	
2 Condovers Cottages Church Road, East Tilbury, RM18 8QX	Diana Mary Finnis Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Jeremy Paul Godsmark Finnis Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Category 31 Potential Claimant under Part 1 of the Land Compensation Act 1973		
Description of Land	Contact Details	
3 Condovers Cottages Church Road, East Tilbury, RM18 8QX	Diana Mary Finnis Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Jeremy Paul Godsmark Finnis Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	
4 Condovers Cottages Church Road, East Tilbury, RM18 8QX	Grace Elizabeth Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	
5 Condovers Cottages Church Road, East Tilbury, RM18 8QX	Grace Elizabeth Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Category 31 Potential Claimant under Part 1 of the Land Compensation Act 1973		
Description of Land	Contact Details	
The Barn Church Road, West Tilbury, RM18 8QU	John Edward Lawrence The Barn Church Road West Tilbury Essex RM18 8QU	
Sunny Side, Church Road, West Tilbury, RM18 8QU	Brian Dean Richardson Sunny Side Church Road West Tilbury Essex RM18 8QU  Mary Louise French Sunny Side Church Road West Tilbury RM18 8QU	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Category 31 Potential Claimant under Part 1 of the Land Compensation Act 1973		
Description of Land	Contact Details	
47 Walnut Tree Cottage, Church Road, West Tilbury, RM18 8QU	Andrew James Osborne 47 Walnut Tree Cottage Church Road West Tilbury Essex RM18 8QU Christine Marilyn Osborne 47 Walnut Tree Cottage Church Road West Tilbury Essex RM18 8QU	
Walnut Tree Farm, Church Road, West Tilbury, RM18 8QU	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sheila Elizabeth Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Category 31 Potential Claimant under Part 1 of the Land Compensation Act 1973		
Description of Land	Contact Details	
Condovers Activities Centre, Church Road, East Tilbury, RM18 8QX	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Sheila Elizabeth Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL  Thurrock District Scout Council Condovers Activities Centre Church Road East Tilbury RM18 8QX	
St James Church, Church Road, West Tilbury, RM18 8UB	Nigel Robert Anderson St James Church The Old Church Church Road West Tilbury Essex RM18 8UB	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled-(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

Category 31 Potential Claimant under Part 1 of the Land Compensation Act 1973		
Description of Land	Contact Details	
West Tilbury Hall, Church Road, West Tilbury, RM18 8UB	Dawn Florence Pledger West Tilbury Hall Church Road West Tilbury Tilbury Essex RM18 8UB	
Malagay Barn, Church Road, West Tilbury. RM18 8UB	David Geoffrey Smith Malagay Barn Church Road West Tilbury Essex RM18 8UB Isama Smith Malagay Barn Church Road West Tilbury Tilbury Tilbury Essex RM18 8UB	

<sup>3.</sup> A person is within Category 3 if the Applicant thinks that, if the order as sought by the application were to be made and fully implemented, the person would or might be entitled(a) as a result of the implementing of the order,
(b) as a result of the order having been implemented, or
(c) as a result of use of the land once the order has been implemented,
to make a relevant claim. See section 57 (4) of the Planning Act 2008.

## Part 3

Number on Plan	Description of Land	Persons enjoying easement or right over land
01/01	New rights over 387.05	Port of Tilbury London Limited
	square metres of land	Leslie Ford House
	being arable field, west of	Tilbury
	Fort Road and south of	Essex
	railway, Tilbury.	RM18 7EH
		(in respect of unilateral notice and beneficiary)
	Freehold title EX966447	
		Bloor Homes Limited
		Ashby Road
		Measham
		Swadlincote Swadlincote
		<del>Derbyshire</del>
		DE12 7JP
		<del>(as beneficiary)</del>
		C H Cole and Sons
		Mill House
		Muckingford Road
		West Tilbury
		<del>Tilbury</del>
		Essex
		RM18 8TP
		<del>(in respect of unilateral notice)</del>
01/04	New rights over 8735.56	AWG Land Holdings Limited
	square metres of land	Lancaster House
	being public road (Fort	Lancaster Way
	Road), private road (Power	Ermine Business Park
	Station Approach Road,	Huntingdon
	Fort Road), north of Tilbury	Cambridgeshire
	Power Station, Tilbury.	PE29 6XU
	. Siles Clausii, Filodiyi	(in respect of rights)

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

**Commented [AP5]:** Title register has beenupdated since submsiion, no longer any part 3 interests.

<sup>4.3:</sup> Book of Reference Parts 1-5

Number	Description of Land	Persons enjoying easement or right over land
on Plan	Freehold title EX932756	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)  Eastern Power Networks plc Newington House 237 Southwark Bridge Road London SE1 6NP (in respect of deed of grant)
01/05	Permanent acquisition of 58.58 square metres of land being arable field and drain, east of Fort Road, Tilbury.  Freehold title EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)
01/06	Permanent acquisition of 432.31 square metres of land being arable field, east	Thurrock Power Limited 1st Floor 145 Kensington Church Street

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
Uli Flali	of Fort Road, Tilbury.  Freehold title EX76273  Leasehold title EX102229	London W8 7LP (as beneficiary)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)
01/07	Permanent acquisition of 115779.30 square metres of land being arable field, east of Fort Road, Tilbury.  Freehold title EX76273  Leasehold title EX102229	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of rights for maintanence)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)
01/08	Permanent acquisition of	Thurrock Power Limited

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
on Plan	3253.04 square metres of land being arable field and drain, east of Fort Road, Tilbury.  Freehold Title EX76273 Leasehold Title EX102229	1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of rights for maintanence)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)
01/09	New rights over 3054.05 square metres of land being grassland, overhead transmission lines and pylons, north of Tilbury Power Station, Tilbury.  Freehold title EX639032 Leasehold title EX487728	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
01/10	New rights over 3185.91 square metres of land being hardstanding at	Ingrebourne Valley Limited Cecil House Foster Street

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
	Tilbury Power Substation, Tilbury.	Harlow Common Harlow Essex CM17 9HY
	Freehold title EX639032 Leasehold title EX487728	(in respect of unilateral notice and beneficiary)
01/11	New rights over 4698.52 square metres of land being private road (unnamed), north of Tilbury Power Station, Tilbury.	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY
	Freehold title EX639032	(in respect of unilateral notice and beneficiary)
01/13	Permanent acquisition of 4400.60 square metres of land being grassland south of Parsonage Common, Tilbury.  Freehold title EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)
	Preenola title EX966447	Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)
		C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number	Description of Land	Persons enjoying easement or right over land
on Plan		
		Essex
		RM18 8TP
		(in respect of unilateral notice and rights of common)
		Jeremy Godsmark Finnis
		Mill House
		Muckingford Road
		West Tilbury
		Tilbury
		Essex
		RM18 8TP
		(in respect of rights of common)
		Ann Louise Cole
		44 St Johns Road
		Writtle
		Chelmsford
		Essex
		CM1 3EB
		(in respect of rights of common)
		Kathryn Ksenia Finnis
		Wyfields Farm
		Blackbush Lane
		Horndon on the Hill
		Stanford-Le-Hope
		Essex
		SS17 8PT
		(in respect of rights of common)
		Sheila Elizabeth Hodson
		Cherry Orchard Farm
		Conways Road
		Orsett

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
		Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
01/14	New rights over 855.70 square metres of land being arable field, overhead transmission lines and drain, Parsonage Common, Tilbury  Freehold title EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)  Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary)
01/15	New rights over 968.66 square metres of land being hardstanding at Tilbury Power Substation, Tilbury.	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
	Freehold title EX639032 Leasehold title EX487728	(in respect of unilateral notice and beneficiary)
01/16	Permanent acquisition of 100695.37 square metres of land being grassland, drains, pylons and overhead transmission lines, south of Parsonage Common, Tilbury.  Freehold title EX966447	Bloor Homes Limited Ashby Road Measham Swadlincote Derbyshire DE12 7JP (as beneficiary) Jeremy Godsmark Finnis
		Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
		Ann Louise Cole 44 St Johns Road Writtle Chelmsford Essex CM1 3EB (in respect of rights of common)
		Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
OII I Iaii		SS17 8PT
		(in respect of rights of common)
		Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL
		(in respect of rights of common)
		Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
		Port of Tilbury London Limited Leslie Ford House Tilbury RM18 7EH (and as beneficiary)
		C H Cole and Sons Mill House Muckingford Road West Tilbury Tilbury Essex
		RM18 8TP
		(in respect of rights of common)

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
01/17	Permanent acquisition of 131997.60 square metres of land being arable field, drains, pylons and overhead transmission lines, south of Parsonage Common, Tilbury.  Freehold title EX76275, EX95363 Leasehold title EX102231	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)
01/19	Permanent acquisition of 11445.30 square metres of land being grassland, hardstanding, private road (unnamed), trees, shrubbery and overhead transmission lines, south east of Tilbury Power Substation, Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)  Hanson Quarry Products Europe Limited Hanson House 14 Castle Hill Maidenhead SL6 4JJ (in respect of rights)
01/20	Permanent acquisition of 27967.4849805.08 square metres of land being	Ingrebourne Valley Limited Cecil House Foster Street

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
	grassland, pylons and overhead transmission lines, east of Walton Common, Tilbury.	Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
	Freehold title EX639032	
01/21	New rights over 98.22 square metres of land being hardstanding, north east of Tilbury Power Station, Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
01/22	Permanent acquisition of 1060.13 square metres of land being hardstanding at Tilbury Power Station, Tilbury.  Freehold title EX639032 Leasehold title EX826359	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
01/25	New rights over 297.70 square metres of land being arable field, drains, south east of Parsonage Common, Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)
		Thurrock Power Limited 1st Floor

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
		145 Kensington Church Street London W8 7LP (as beneficiary)
01/27	Permanent acquisition of 10460.75 square metres of land being arable field, south east of Walton Common, Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
01/28	Permanent acquisition of 1992.46 square metres of land being grassland east of Tilbury Power Station.  Freehold title EX639032	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)
01/30	New rights over 7490.22 square metres of land being grassland, hardstanding, private road (unnamed), trees, shrubbery and overhead transmission lines, south east of Tilbury Power	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
	Substation, Tilbury.	Hanson Quarry Products Europe Limited
	Freehold title EX639032	Hanson House 14 Castle Hill Maidenhead SL6 4JJ (in respect of rights)
01/31	New rights over 95.22 square metres of land being hardstanding and private road east of Walton Common, Tilbury.	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex
	Freehold title EX639032	CM17 9HY (in respect of unilateral notice and beneficiary)
01/32	Permanent acquisition of 52.12 square metres of land hardstanding, east of Walton Common, Tilbury.  Freehold title EX639032	Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
02/01	New rights over 143.18 square metres of land being arable field and drain, north west of Parsonage Common, Tilbury.	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary)
	Freehold title EX966447	Bloor Homes Limited Ashby Road Measham

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
01111011		Swadlincote
		Derbyshire
		DE12 7JP
		(as beneficiary)
		C H Cole and Sons
		Mill House
		Muckingford Road
		West Tilbury
		Tilbury
		Essex
		RM18 8TP
		(in respect of rights of common)
		Jeremy Godsmark Finnis
		Mill House
		Muckingford Road
		West Tilbury
		Tilbury Essex
		RM18 8TP
		(in respect of rights of common)
		(in respect of rights of common)
		Ann Louise Cole
		44 St Johns Road
		Writtle
		Chelmsford
		Essex
		CM1 3EB
		(in respect of rights of common)
		Kathryn Ksenia Finnis
		Wyfields Farm
		Blackbush Lane

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
On Figure		Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
02/02	New rights over 6964.99 square metres of land being arable field and drain, west of Parsonage Common, Tilbury.  Freehold title EX76273 Leasehold title EX102229	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Barking Power Limited Guildhall PO BOX 270 London

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
		EC2P 2EJ (in respect of deed)
02/03	Permanent acquisition of 10596.91 square metres of land being arable field and drain, east of Fort Road, Tilbury.  Freehold title EX76273  Leasehold title EX102229	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Thurrock Borough Council Civic Centre New Road Grays Essex RM17 6SL (in respect of rights for maintanence)  Barking Power Limited Guildhall PO BOX 270 London EC2P 2EJ (in respect of deed)
02/04	New rights over 1539.37 square metres of land being Parsonage Common, south of Cooper Shaw Road, Tilbury  Freehold title EX966447	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary) Bloor Homes Limited Ashby Road

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number	Description of Land	Persons enjoying easement or right over land
on Plan		
		Measham
		Swadlincote
		Derbyshire
		DE12 7JP
		(as beneficiary)
		C H Cole and Sons
		Mill House
		Muckingford Road
		West Tilbury
		Tilbury
		Essex
		RM18 8TP
		(in respect of rights of common)
		Jeremy Godsmark Finnis
		Mill House
		Muckingford Road
		West Tilbury
		Tilbury
		Essex
		RM18 8TP
		(in respect of rights of common)
		Ann Louise Cole
		44 St Johns Road
		Writtle
		Chelmsford
		Essex
		CM1 3EB
		(in respect of rights of common)
		Kathryn Ksenia Finnis
		Wyfields Farm

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number	Description of Land	Persons enjoying easement or right over land
on Plan		Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
02/05	Permanent acquisition of 78.25 square metres of land being trees at Parsonage Common, West Tilbury.  Freehold title EX95363	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)
02/06	Permanent acquisition of 45291.69 square metres of	Thurrock Power Limited 1st Floor

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
	land being arable field, track, drains, pylons and overhead transmission lines, south of Parsonage Common, West Tilbury.  Freehold title EX76275, EX95363 Leasehold title EX102231	145 Kensington Church Street London W8 7LP (as beneficiary)  RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)
02/07	Permanent acquisition of 52640.80 square metres of land being arable field, shrubbery, drains, pylons, overhead transmission lines at Parsonage Shaw, West Tilbury.  Freehold title EX95363, EX418108 Leasehold title EX102231	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
02/09	New rights over 25.27 square metres of land being grassland and drain at Parsonage Common, West Tilbury.  Unregistered Leasehold title EX102231	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
02/11	New rights over 253.28 square metres of land being shrubbery at Parsonage Shaw, West Tilbury.  Freehold title EX95363, EX418108 Leasehold title EX102231	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
02/12	Permanent acquisition of 2376.65 square metres of land being grassland, track and drain, south of Parsonage Shaw, West Tilbury.  Unregistered	Diana Mary Cole Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of way)  James Andrew Cole Mill House Muckingford Road West Tilbury Tilbury RM18 8TP (in respect of rights of way)
02/13	New rights over 60035.74 square metres of land being arable field, track,	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
	drains, pylons and overhead transmission lines, west of Parsonage Shaw, West Tilbury.  Freehold title EX246891	Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
02/14	Permanent acquisition of 38353.62 square metres of land being arable field, track and drains, west of Parsonage Shaw, West Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
03/01a	New rights over 7893.09 square metres of land being arable land, pylons and overhead transmission lines and drain, south of Station Road, Tilbury.	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
on Plan	Freehold title EX246891	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
03/01	New rights over 34276.76 square metres of land being arable land, pylons and overhead transmission lines and drain, south of Station Road, Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)
03/02	New rights over 5690.37 square metres of land being arable land and drain, south of Station Road, Tilbury.  Freehold title EX246891	RWE Generation (UK) plc Windmill Hill Business Park Whitehill Way Swindon Wiltshire SN5 6PB (as beneficiary)  Thurrock Power Limited 1st Floor 145 Kensington Church Street

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
		London W8 7LP (as beneficiary)
03/03	New rights over 1168.17 square metres of land being public footpath (Footpath 146), trees and drain, south west of Station Road, East Tilbury.  Freehold title EX972747	Thurrock Power Limited 1st Floor 145 Kensington Church Street London W8 7LP (as beneficiary)  Cogent Land LLP 33 Margaret Street London W1G 0JD (in respect of an option)  International Power Limited Level 20 25 Canada Square London E14 5LQ (in respect of deed)  Jeremy Godsmark Finnis Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
		Ann Louise Cole

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
on Plan		44 St Johns Road Writtle Chelmsford Essex CM1 3EB (in respect of rights of common)  Kathryn Ksenia Finnis Wyfields Farm Blackbush Lane Horndon on the Hill Stanford-Le-Hope Essex SS17 8PT (in respect of rights of common)  Sheila Elizabeth Hodson Cherry Orchard Farm Conways Road Orsett Grays RM16 3EL (in respect of rights of common)  Sue Cole Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
		C H Cole and Sons

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
		Mill House Muckingford Road West Tilbury Tilbury Essex RM18 8TP (in respect of rights of common)
03/04	New rights over 14664.93 square metres of land being agricultural field, trees and shrubbery, west of Gravelpit Farm, Tilbury.  Freehold title EX246889	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)  International Power Limited Level 20 25 Canada Square London E14 5LQ (in respect of deed)
03/06	New rights over 15027.72 square metres of land being arable field, grassland, trees and shrubbery at Gravelpit Farm, Tilbury.  Freehold title EX246889 and EX763684	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)
03/07	Temporary rights over 1167.65 square metres of land being private road	Melville Hamilton Lowe Mott Goshem's Farm Station Road

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
OH Hall	(unnamed), west of Goshem's Farm, East	East Tilbury Tilbury
	Tilbury.	Essex RM18 8QR
	Unregistered	(in respect of rights of access)
		Rita Maureen Mott Goshem's Farm Station Road East Tilbury Tilbury RM18 8QR (in respect of rights of access)
		David Glenn Mott Norrsken Station Road East Tilbury Tilbury RM18 8QR (in respect of rights of access)
		Karen Misra Norrsken Station Road East Tilbury Tilbury RM18 8QR (in respect of rights of access)
		Exel Logistics Property Limited Ocean House The Ring Bracknell

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
		Berkshire RG12 1AN (in respect of rights of access)
03/09	New rights over 21164.86 square metres of land being grassland and arable field at Goshem's Farm, East Tilbury.  Freehold title EX246889	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary)
03/11	Permanent acquisition of 22827.83 square metres of land being agricultural and arable fields, trees and shrubbery at Goshem's Farm, East Tilbury.  Freehold title EX246889	Cogent Land LLP 33 Margaret Street London W1G 0JD (as beneficiary) International Power Limited Level 20 25 Canada Square London E14 5LQ (in respect of deed)
04/01	New rights over 7181.29 square metres of land being private access road (unnamed), hardstanding, grassland and shrubbery, east of Tilbury Power Station, Tilbury.  Freehold title EX639032	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (as beneficiary) Ingrebourne Valley Limited Cecil House

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
		Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
04/02	Permanent acquisition of 15213.50 square metres of land being public footpath (Footpath 146), grassland, trees, shrubbery and hardstanding, east of Tilbury Power Station, Tilbury.  Freehold title EX639032	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)  Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)
04/03	New rights over 11346.49 square metres of land being private acces road (unnamed), quarryland and grassland, east of Tilbury Power Station.  Freehold title EX639032	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)  Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

Number on Plan	Description of Land	Persons enjoying easement or right over land
		(in respect of unilateral notice and beneficiary)
04/05	Permanent acquisition of 7692.92 square metres of land being grassland, east of Tilbury Power Station, Tilbury.  Freehold title EX639032	Port of Tilbury London Limited Leslie Ford House Tilbury Essex RM18 7EH (in respect of unilateral notice and beneficiary)  Ingrebourne Valley Limited Cecil House Foster Street Harlow Common Harlow Essex CM17 9HY (in respect of unilateral notice and beneficiary)

<sup>4.</sup> Part 3 contains the names of all those entitled to enjoy easements or other private rights over land (including private rights of navigation over water) which it is proposed shall be extinguished, suspended or interfered with. See regulation 7 (1)(c) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

## Part 4<sup>5</sup>

Number on Plan	Description of Land	Freehold Owners or Reputed Freehold Owners	Other owners
None	None	None	None

<sup>5.</sup> Part 4 specifies the owner of any Crown interest in the land which is proposed to be used for the purposes of the order for which application is being made. See regulation 7 (1)(d) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

## Part 56

Number on Plan	Description of Land	Category of land
01/07	Permanent acquisition of 115779.30 square metres of land being arable field, east of Fort Road, Tilbury.  Freehold title EX76273 Leasehold title EX102229	Replacement Common Land
01/13	Permanent acquisition of 4400.60 square metres of land being grassland south of Parsonage Common, Tilbury.  Freehold title EX966447	Common Land
01/14	New rights over 855.70 square metres of land being arable field, overhead transmission lines and drain, Parsonage Common, Tilbury  Freehold title EX966447	Common Land
01/16	Permanent acquisition of 100695.37 square metres of land being grassland, drains, pylons and overhead transmission lines, south of Parsonage Common, Tilbury.	Common Land

<sup>6.</sup> Part 5 specifies land(i) the acquisition of which is subject to special parliamentary procedure;
(ii) which is special category land;
(iii) which is replacement land. See regulation 7 (1)(e) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

4.3: Book of Reference Parts 1-5

143

Name of the American Control o		Ottomarafland
Number on Plan	Description of Land	Category of land
on Plan		
	Freehold title EX966447	
02/01	New rights over 143.18 square metres of land being arable field and drain, north west of Parsonage Common, Tilbury.  Freehold title EX966447	Common Land
02/04	New rights over 1539.37 square metres of land being Parsonage Common, south of Cooper Shaw Road, Tilbury  Freehold title EX966447	Common Land
03/03	New rights over 1168.17 square metres of land being public footpath (Footpath 146), trees and drain, south west of Station Road, East Tilbury  Freehold title EX972747	Common Land

<sup>6.</sup> Part 5 specifies land(i) the acquisition of which is subject to special parliamentary procedure;
(ii) which is special category land;
(iii) which is replacement land. See regulation 7 (1)(e) of the Infrastructure Planning (Applications: Prescribed Forms and Procedure) Regulations 2009

4.3: Book of Reference Parts 1-5